Author's response to reviews

Title: LEARN 2 MOVE 16-24: Effectiveness of an intervention to stimulate physical activity and improve physical fitness of adolescents and young adults with spastic cerebral palsy; a randomized controlled trial

Authors:

Jorrit Slaman (j.slaman@erasusmc.nl)
Marij Roebroeck (m.roebroeck@erasusmc.nl)
Jetty van Meeteren (j.vanmeeteren@erasusmc.nl)
Wilma van der Slot (w.vanderslot@erasusmc.nl)
Heleen Reinders-Messelin (H.A.Reinders@revalidatie-friesland.nl)
Eline Lindeman (e.lindeman@umcutrecht.nl)
Henk Stam (h.j.stam@erasusmc.nl)
Rita van den Berg-Emons (h.j.g.vandenberg@erasusmc.nl)

Version: 3 Date: 23 August 2010

Author's response to reviews: see over
Subject: Submission of a manuscript for evaluation

Dear Editor,

I am enclosing herewith the revised manuscript entitled “LEARN 2 MOVE 16-24: Effectiveness of an intervention to stimulate physical activity and improve physical fitness of adolescents and young adults with spastic cerebral palsy; a randomized controlled trial” for publication in “BMC Pediatrics”.

Changes in the manuscript:
- The opening paragraph (which was common in all protocols) is deleted.
- Some descriptive information about CP has been added to the beginning of the background section
- References to the other Learn2Move manuscripts are placed at the end of the background section.

Sincerely,

Jorrit Slaman
Erasmus MC, University Medical Center Rotterdam
PO Box 2040, 3000 CA Rotterdam, The Netherlands
Email: j.slaman@erasmusmc.nl
Phone: +31-10-7044601
Fax: +31-10-7033843